Author’s response to reviews

Title: Multimodality curative treatment of salivary gland cancer liver metastases with drug-eluting beads chemoembolization, radiofrequency ablation and surgical resection: A case report

Authors:

Andreas Karatzas (andkaratzas@yahoo.co.uk)
Konstantinos Katsanos (katsanoskonstantinos@gmail.com)
Ioannis Maroulis (ioan.maroulis@gmail.com)
Christina Kalogeropoulou (ckrat@upatras.gr)
Evaggelos Tzorakoleftherakis (etzorakol@med.upatras.gr)
Dimitris Karnabatidis (karnaby@upatras.gr)

Version: 3 Date: 30 June 2011

Author’s response to reviews: see over
Dear Editor,

According to your comments concerning our manuscript titled: “Multimodality curative treatment of salivary gland cancer liver metastases with drug-eluting beads chemoembolization, radiofrequency ablation and surgical resection: A case report”. We have revised the manuscript according to the reviewers’ suggestions. Please find below a point-by-point list of all the changes made. Underlined phrases denote amended or newly added text in the revised manuscript.

Sincerely,

The authors,

Andreas Karatzas
Konstantinos Katsanos
Ioannis Maroulis
Christina Kalogeropoulou
Evaggelos Tzorakoleftherakis
Dimitris Karnabatidis
1) We have added the race and ethnicity of the patient both in the abstract and case presentation, adding in both the phrase white, Greek.

2) We have added at the case presentation a phrase to make clear that the patient was asymptomatic and another phrase to describe the patient’s past medical history. You can find in the 1st paragraph of the case presentation the following: The patient was asymptomatic and physical examination was unremarkable. Routine laboratory examinations showed no abnormal findings. Apart from the history of large bowel polypectomies she only had a history of appendectomy 35 years ago and bilateral saphenectomy 23 years ago.

3) We have removed the additional file with the annotated text

4) We have removed the text from the corners of figures 1b, 1c.